Author's response to reviews

Title: "Beware of red herrings in the hip clinic!" - an unusual case of persistent groin pain after total hip arthroplasty: a case report

Authors:

Praveen Konala (p.konala@imperial.ac.uk)
Schaefer K Thomas (royalts@hotmail.com)
Iranpour Farhad (f.iranpour@imperial.ac.uk)
Niklaus F Friederich (niklaus.friederich@ksbh.ch)
Michael T Hirschmann (Michael.Hirschmann@ksbh.ch)

Version: 3 Date: 18 October 2010

Author's response to reviews:

Dear Professor Kidd,

thank you very much for the thorough review and giving us the opportunity to resubmit our manuscript entitled “Beware of red herrings in the hip clinic! – an unusual case of persistent groin pain after total hip arthroplasty” from the Department of Orthopaedic Surgery and Traumatology, Kantonsspital Bruderholz, Switzerland by Konala, Schaefer, Iranpour, Friederich and Hirschmann.

We have addressed all the reviewer comments in a point by point response to facilitate further review.

The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. We declare that all authors have contributed to the paper and are familiar with the contents of the final draft. We hereby affirm that the submitted manuscript is our original work, has not been published or is being considered for publication elsewhere. There is no conflict of interest. Looking forward to hearing from you at your earliest convenience I remain,

Yours sincerely,

Michael Hirschmann, MD

Reviewer comments

The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. We declare that all authors have contributed to the paper and are familiar with the contents of the final draft. We hereby affirm that the submitted manuscript is our original work, has not been published or is being considered for publication elsewhere. There is no conflict of interest. Looking forward to hearing from you at your earliest convenience I remain,
Yours sincerely,  
Michael Hirschmann, MD

Reviewer comments  
The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. We declare that all authors have contributed to the paper and are familiar with the contents of the final draft. We hereby affirm that the submitted manuscript is our original work, has not been published or is being considered for publication elsewhere. There is no conflict of interest. Looking forward to hearing from you at your earliest convenience I remain,  
Yours sincerely,  
Michael Hirschmann, MD

Reviewer comments  
The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. We declare that all authors have contributed to the paper and are familiar with the contents of the final draft. We hereby affirm that the submitted manuscript is our original work, has not been published or is being considered for publication elsewhere. There is no conflict of interest. Looking forward to hearing from you at your earliest convenience I remain,  
Yours sincerely,  
Michael Hirschmann, MD

Reviewer comments  
The final manuscript has been seen and approved by all authors and they have taken due care to ensure the integrity of the work. We declare that all authors have contributed to the paper and are familiar with the contents of the final draft. We hereby affirm that the submitted manuscript is our original work, has not been published or is being considered for publication elsewhere. There is no conflict of interest. Looking forward to hearing from you at your earliest convenience I remain,  
Yours sincerely,  
Michael Hirschmann, MD
interest. Looking forward to hearing from you at your earliest convenience I remain,

Yours sincerely,
Michael Hirschmann, MD

Reviewer comments
The abstract has been restructured as suggested.
The manuscript was once again checked and edited by a native english speaker.